Author's response to reviews

Title: Long-term safety of Mometasone Furoate administered via a dry powder inhaler in children: Results of an open-label study comparing Mometasone Furoate with Beclomethasone Dipropionate in children with persistent asthma

Authors:

Michael Noonan (allergy1@asthma-research.com)
Jeffrey Leflein (jlefein@aiaofaa.com)
Jonathan Corren (jc@allergymedicalclinic.com)
Heribert Staudinger (Heribert.Staudinger@spcorp.com)

Version: 4 Date: 4 July 2009

Author's response to reviews: see over
July 2, 2009

Melissa Norton, MD
Editor-in-Chief, *BMC Pediatrics*

Dear Dr. Norton,

On behalf of all the authors, I would like to thank *BMC Pediatrics* for your acceptance of our manuscript entitled “*Long-Term Safety of Mometasone Furoate Administered via a Dry Powder Inhaler in Children: Results of an Open-Label Study Comparing Mometasone Furoate With Beclomethasone Dipropionate in Children With Persistent Asthma*” on which I am the lead and corresponding author. We have made the formatting corrections on the most recent version of the article as requested by *BMC Pediatrics*.

Regarding your request in reference to Figure 3, the Figure numbers listed previously on page 9 were incorrect (3A, 3B) and have been corrected (2A, 2B). In addition, a grammatical error on page 8 and a typographical error on page 14 have been corrected. Also, reference numbers 1, 2, and 8 (on page 16) have been corrected and the EndNote field codes have been removed from the document file.

We look forward to final online publication of the article.

Sincerely,

Michael Noonan, MD
Allergy Associates Research Center
545 NE 47th Avenue, Suite 310
Portland, OR 97213
Phone: 503-238-6233
Fax: 503-231-7668
allergy1@asthma-research.com